Author's response to reviews

Title: Impaired psychological recovery in the elderly after the Niigata-Chuetsu Earthquake in Japan: a population-based study

Authors:

Shin-ichi Toyabe (toyabe@med.niigata-u.ac.jp)
Toshiki Shioiri (tshioiri@med.niigata-u.ac.jp)
Hideki Kuwabara (hkuwa@med.niigata-u.ac.jp)
Toshiyuki Someya (someya@med.niigata-u.ac.jp)
Kouhei Akazawa (akazawa@medws1.med.niigata-u.ac.jp)
Nohito Tanabe (tanabe@med.niigata-u.ac.jp)

Version: 4 Date: 13 September 2006

Author's response to reviews:

Dear Editor,

Thank you for your E-mail of September 13. I have changed the formatting of the table and figures according to your advice.

Sincerely,

Shin-ichi Toyabe, MD